Author's response to reviews

Title: Health sector reforms and human resources for health in Uganda and Bangladesh: mechanisms of effect

Authors:

Freddie F Ssengooba (Freddie.Ssengooba@lshtm.ac.uk)
Syed A Rahman (Aziz.Rahman@lshtm.ac.uk)
Charles C Hongoro (charles.hongoro@lshtm.ac.uk)
Elizeus E Rutebemberwa (rutebemberwa@yahoo.com)
Ahmed A Mustafa (dramheu@yahoo.co.uk)
Tara T Kielmann (tara.kielmann@ed.ac.uk)
Barbara B McPake (BMcPake@qmuc.ac.uk)

Version: 2 Date: 26 January 2007

Author's response to reviews:

Editorial changes.